Author's response to reviews

Title: Effectiveness of a Primary care-based intervention to Reduce Sitting Time in Overweight and Obese Patients (SEDESACTIV). A Randomized Controlled Trial. Rationale and study design

Authors:

Carme Martín-Borràs (mariacarmenmb@blanquerna.url.edu)
Maria Giné-Garriga (mariagg@blanquerna.url.edu)
Elena Martínez-Ramos (elenamar1710@gmail.com)
Carlos Martín-Cantera (carlos.martin@uab.es)
Elisa Puigdoménech (cardiocat@gmail.com)
Mercè Solà-Gonfaus (msola@ambitcp.catsalut.net)
Eva Castillo (evacastillo_58@hotmail.com)
Anna Puig-Ribera (annam.puig@uvic.cat)
Angela Mª Beltrán (marian2081@yahoo.com)
José Manuel Trujillo (aplaceinheaven@hotmail.com)
Beatriz Rodríguez (beatriz_rodriguez_roca@fremap.es)
Olga Pueyo (olgapueyo55@gmail.com)
Javier Pueyo (pueyo.javiel@gmail.com)
Noemí Serra (nserra@inefc.es)
SEDESTACTIV Study Group (sedestactiv@gmail.com)

Version: 2
Date: 7 February 2014

Author's response to reviews: see over
7th February 2014

Dear,

Regarding our manuscript entitled ‘Effectiveness of a Primary care-based intervention to Reduce Sitting Time in Overweight and Obese Patients (SEDESTACTIV). A Randomized Controlled Trial. Rationale and study design’ (MS: 7886277331128309), I’m delighted to inform you that I provide the requested responses:

1. **Ethical and Funding Approval Documentation.** I have forwarded the requested documentation to HYPERLINK "mailto: BMCSeriesEditorial@biomedcentral.com". I have included: ethical approval (english version and the original document) and funding approval (this document is in Spanish, but we could request english version if you consider that is necessary).

2. **Funding.** As we had included in the Acknowledgements section of our manuscript, the authors have been awarded a grant for the study by *Instituto de Salud Carlos III* (PI11/01082). If you consider we could modify the section’s title (“Acknowledgments” by “Funding and Acknowledgements”).

3. **Study status.** SEDESTACTIV is an ongoing study where the investigators are still collecting.

4. **Related Articles.** The work is original and has not been submitted elsewhere for publication, we hope that you will find it suitable for publication in BMC Public Health.

5. We have also updated the: (a) URL of register in the Manuscript details, and (b) Figure1 in ‘Upload Files’ section.

Thank you very much for your attention and for further information don’t hesitate to contact me.

Sincerely,

Carme.

Carme Martín-Borràs,
FPCEE Blanquerna
Universitat Ramon Llull